

Format for ANSWERING REVIEWERS

August 16/08/2016



Title: Hepatic metastases from gastric cancer. Are we ready for the challenge? A Surgical Perspective

Author: Guido A.M. Tiberio, Franco Roviello, Annibale Donini, Giovanni de Manzoni, on behalf of the Italian Research Group for Gastric Cancer

Name of Journal: *World Journal of Gastroenterology*

ESPS Manuscript NO: 19810

Response Letter

Dear Editor,

The manuscript has been improved following your instructions:

- The text was submitted to and revised by AJE for English language editing; an editorial certificate is attached
- The paper was cross-checked and results are in an attached file
- Comment 1 was accepted and the title revised accordingly
- A short running title was prepared
- Comment 3 was accepted
- Authors' contributions were recorded
- Conflict of interest declared
- Comment 6 was accepted
- Comment 7 was accepted and key-words listed
- Core tip was inserted (Comment 8)
- Audio Core tip is in an attached file
- Reference numbers are in square brackets
- Reference list was expanded to 25 citations, not to 30 because we did not find adequate papers to be listed in an editorial' bibliography.
- There were no specific suggestions or requests from the Reviewers

Thank you again for publishing our manuscript in the *World Journal of Gastroenterology*.

Sincerely yours,

Guido A.M. Tiberio, Prof. of Surgery
University of Brescia